Author's response to reviews

Title: The efficacy of computer reminders on external quality assessment for point-of-care testing in Danish general practice: Rationale and methodology

Authors:

Frans B Waldorff (fransw@sund.ku.dk)
Volkert Siersma (siersma@sund.ku.dk)
Ruth Ertmann (ruth@sund.ku.dk)
Marius Brostrøm Kousgaard (marbro@sund.ku.dk)
Annette Sonne Nielsen (anette.sonne.nielsen@regionh.dk)
Peter Felding (pf@kpll.dk)
Niels Mosbæk (nmo@kpll.dk)
Else Hjortso (else.hjortsoe@regionh.dk)
Susanne Reventlow (susrew@sund.ku.dk)

Version: 2 Date: 15 February 2011

Author's response to reviews:

Dear editor,

I tried to upload the notification from the ethics committee but I could not because your system does not accept pdf files. I have attached this document in a separate mail to Ian Cauton. The committee evaluated the project and decided that this project was not a biomedical research project. Therefore, the project is not notifiable according to Danish law and the project can proceed without the approval of Ethics Committee.

The project has not received any external grant but is conducted as part at my job as an associate research professor and therefore I can not provide you with a specific funding letter. The resources used on the electronic reminder system came from the laboratory as part of a internal strategy in order to improve adherence to guidelines. The qualitative interviews are conducted by a staff member of a regional quality improvement unit as part of her employment.

Best wishes
Frans Boch Waldorff